Author’s response to reviews

Title: PHYSICAL ACTIVITY AND FITNESS ARE ASSOCIATED WITH VERBAL MEMORY, QUALITY OF LIFE AND DEPRESSION AMONG NURSING HOME RESIDENTS: PRELIMINARY DATA OF A RANDOMIZED CONTROLLED TRIAL

Authors:

Haritz Arrieta (haritz.arrieta@ehu.eus)
Chloe Rezola (chloe.rezola@ehu.eus)
Iñaki Echeverria (inaki.echeverriag@ehu.eus)
Miren Iturburu (miren.iturburu@matiainstituto.net)
Susana Maria Gil (susana.gil@ehu.eus)
Jose Javier Yanguas (javiyanguaslezaun@gmail.com)
Jon Irazusta (jon.irazusta@ehu.eus)
Ana Rodriguez Larrad (ana.rodriguez@ehu.eus)

Version: 2 Date: 08 Dec 2017

Author’s response to reviews:

Dear Dr. Anne-Marie Hill,

Thank you for the comments raised from the review process of the manuscript “PHYSICAL ACTIVITY AND FITNESS ARE ASSOCIATED WITH VERBAL MEMORY, QUALITY OF LIFE AND DEPRESSION AMONG NURSING HOME RESIDENTS: PRELIMINARY DATA OF A RANDOMIZED CONTROLLED TRIAL” (BGTC-D-17-00526). We really appreciate the thoughtful critiques and believe they have allowed us to improve the manuscript. We have fully addressed the comments indicated by the editor and reviewers and made the necessary corrections in a revised version of the manuscript (all changes are indicated in the text by highlighting). The point-by-point rebuttal follows in an attached document. We hope that the revised manuscript will be favorably considered for publication in BMC Geriatrics.
Sincerely,

Ana Rodriguez Larrad